Author's response to reviews

Title: Protocol of an expertise based randomized trial comparing surgical Venae Sectio versus radiological Puncture of Vena Subclavia for insertion of Totally Implantable Access Port in oncological patients (PORTAS 2 Trial: NCT 006600444)

Authors:

Phillip Knebel (phillip.knebel@med.uni-heidelberg.de)
Lars Fischer (lars.fischer@med.uni-heidelberg.de)
Eva Cremonese (evacremonese@hotmail.com)
Ruben Lopez-Benitez (ruben.lopez@med.uni-heidelberg.de)
Ulrike Stampfl (ulrike.stampfl@med.uni-heidelberg.de)
Boris Radeleff (boris.radeleff@med.uni-heidelberg.de)
Hans-Ulrich Kauczor (hans-ulrich.kauczor@med.uni-heidelberg.de)
Markus W Büchler (markus.buechler@med.uni-heidelberg.de)
Christoph M Seiler (christoph.seiler@med.uni-heidelberg.de)

Version: 3 Date: 13 October 2008

Author's response to reviews:

Dear Reviewer,

1. Corrected as recommended
2. Corrected as recommended
3. Corrected as recommended
4. Corrected as recommended
5. Corrected as recommended
6. Corrected as recommended
7. Corrected as recommended
8. Corrected as recommended
9. Corrected as recommended

With kind regards and greetings
Phillip Knebel